Author's response to reviews

Title: Interaction among general practitioners age and patient load in the prediction of job strain, decision latitude and perception of job demands

Authors:

Giedrius Vanagas (vanagas@kmu.lt)
Susanna Bihari-Axelsson (susaxe@nhv.se)

Version: 4 Date: 19 October 2004

Author's response to reviews:

Dear Editor and Reviewers,

Thank you very much for very valuable and correct comments and suggestions regarding our article. We followed your suggestions:

1. Descriptions of results in summary made more informative and oriented at the meaning of results.
2. Part of discussion material posed to introduction.
3. Last paragraphs of introduction are our research question and aim.
4. Explained why we expect that age and patient load influences the work characteristics (4th paragraph).
5. Made more understandable descriptions of statistics in results section.
6. Rewritten part of discussion (added analysis of weaknesses and strength) regarding to the suggestions.
7. Corrected reference list (e.g. ref 7).

Hopefully we met your expectations
Thank you again for very valuable suggestions,
Authors